Author's response to reviews

Title: MTHFR C677T and MTR A2756G Polymorphisms and the Homocysteine Lowering Efficacy of Different Doses of Folic Acid in Hypertensive Chinese Adults

Authors:

Xianhui Qin (xianhuiqin@gmail.com)
Jianping Li (lijianpingbi@163.com)
Yimin Cui (cuiymzy126@126.com)
Zeyuan Liu (lzy13311010528@tom.com)
Zhigang Zhao (tt1022zzg@163.com)
Junbo Ge (gejunbosh@126.com)
Deming Guan (yaoliiji_guan@163.com)
Jian Hu (hujian913@yeah.net)
Yanni Wang (ynwang78@yeah.net)
Fumin Zhang (Fmzhang88@126.com)
Xin Xu (xinxxu@126.com)
Xiaobin Wang (xiaobinwang18@gmail.com)
Xiping Xu (pharmaqxh@gmail.com)
Yong Huo (huoyong18@126.com)

Version: 3 Date: 6 January 2012

Author's response to reviews:

Dear Editor:

We are sure that the title is identical in the manuscript and the submission system.

We have removed all highlighting and tracking from the manuscript and made no change for the manuscript.

Please do not hesitate to contact us if we can be of any further assistance.

Many thanks.

Sincerely,
Xiping Xu, MD, PhD
Institute of Biomedicine, Anhui Medical University, Hefei, China
Email: pharmaqxh@gmail.com
Phone: +86-551-5161211
Fax: +86-551-5161211